\documentclass [12pt] {article}

\usepackage{graphicx, xypic,amsmath}

\topmargin 0pt
\advance \topmargin by -\headheight
\advance \topmargin by -\headsep
\textheight 8.9in
\oddsidemargin 0pt
\evensidemargin \oddsidemargin
\marginparwidth 0.5in
\textwidth 6.5in

\begin{document}
\begin{center}

\Huge{
The Unit Circle}
\end{center}

\normalsize The circle below is the unit circle with the terminal sides of ``common" angles marked.  Common angles correspond to special triangles we will be working with today \& Thursday and are multiples of $45^\circ$ and $30^\circ$. For example, the point $D$ corresponds to 60 degrees or $\frac{\pi}{3}$ radians.  Fill in the following chart

\vskip -1.5in
\hskip 2.2in \includegraphics[height=7in]{blankunitcircle}


\vskip -1.5in




\vskip -3.5in  \begin{tabular*}{3in}{l @{\extracolsep{\fill}} l @{\extracolsep{\fill}} l}
point &degree & radian  \\
\\
A &&\\
\\
\hline
B&&\\
\\
\hline
C&&\\
\\
\hline
D&60&$\displaystyle{\frac{\pi}{3}}$\\
\\
\hline
E&&\\
\\
\hline
F&&\\
\\
\hline
G&&\\
\\
\hline
H&&\\
\\
\hline
I&&\\
\\
\hline
J&&\\
\\
\hline
K&&\\
\\
\hline
L&&\\
\\
\hline
M&&\\
\\
\hline
N&&\\
\\
\hline
O&&\\
\\
\hline
P&&\\
\end{tabular*}

\newpage
\includegraphics[height=8.5in]{WebUnitCircle}


\end{document}